Author's response to reviews

Title: Cost-minimization analysis of the direct costs of TPE and IVIg in the treatment of Guillain-Barre syndrome

Authors:

Jeffrey L Winters (winters.jeffrey@mayo.edu)
David Brown (DBrown1@hne.com)
Elisabeth Hazard (hodiehazard@yahoo.com)
Ashok Chainani (Ashok.Chainani@caridianbct.com)
Chester Andrzejewski Jr (chester.andrzejewski@baystatehealth.org)

Version: 3 Date: 15 February 2011

Author's response to reviews:

Dear Dr Galbraith,

After submitting our revision on 2/13/11, it was noted that the abstract had not been updated to include the revised numbers using the more recent data. This has now been done.

Sincerely,

Jeff